Author's response to reviews

Title: Preventive and clinical care provided to adolescents attending New South Wales Public Oral Health Services, Australia: A retrospective study.

Authors:

Angela V Masoe (Angela.Masoe@gsahs.health.nsw.gov.au)
Anthony S Blinkhorn (anthony.blinkhorn@sydney.edu.au)
Jane Taylor (jane.taylor@newcastle.edu.au)
Fiona A Blinkhorn (fiona.blinkhorn@newcastle.edu.au)

Version: 19 Date: 19 October 2014

Author's response to reviews: see over
18th October, 2014

Dear Editor

Thank you for the opportunity to resubmit our manuscript. The authors have read the Reviewers’ comments and have updated the manuscript accordingly.

Reviewers comment: Line 93, page 8, please define the abbreviation LHD the first time it is used.

Response: We have inserted definition for LHD as requested on page 93.

Yours Sincerely

Angela Masoe

Oral Health Researcher

18th October, 2014